
```
Mauna Loa Solar Observatory Observer's Log
    ______
       Wed Jan 3 17:41:05 GMT 2001
Year: 01 Doy: 003
Observer: koon
WEATHER COMMENT: Wed Jan 3 17:42:21 GMT 2001
Thin to medium cirrostratus overcast, wind=5 mph from the SSW.
Wed Jan 3 17:44:06 GMT 2001
                             CHIP
                                     Startup--Initializing new tape
COMMENT: Wed Jan 3 17:52:41 GMT 2001
Spar is barely tracking.
Wed Jan 3 17:52:46 GMT 2001
                             CHIP
                                     Start 7 Passband Patrol
Wed Jan 3 17:52:51 GMT 2001
                            PICS
                                     Start Patrol
Wed Jan 3 18:03:52 GMT 2001
                                     Flat
                            PICS
Wed Jan 3 18:07:18 GMT 2001
                            PICS
                                     End Flat
Wed Jan 3 18:17:40 GMT 2001
                            MKIV
                                    Start Patrol
MKIV COMMENT: Wed Jan 3 19:00:40 GMT 2001
Pausing due to clouds.
ECHO COMMENT: Wed Jan 3 19:00:57 GMT 2001
Adjusted MOF2 temp from 138.0 (default) to 130.0 as requested by Kim and Steve
in an attempt to see if doppler shift improves on the images.
WEATHER COMMENT: Wed Jan 3 19:58:32 GMT 2001
Clouds are much thicker now, spar is barely tracking.
Wed Jan 3 20:01:30 GMT 2001
                            CHIP
                                     LSD
Wed Jan 3 20:04:17 GMT 2001
                             CHIP
                                     End LSD
Wed Jan 3 20:04:27 GMT 2001
                                     BiasLSD
                            CHIP
Wed Jan 3 20:05:12 GMT 2001
                                     End BiasLSD
                            CHIP
Wed Jan 3 20:05:21 GMT 2001
                                     Bias7
                             CHIP
Wed Jan 3 20:06:08 GMT 2001
                            CHIP
                                     End Bias7
**ECHO PROBLEM**: Wed Jan 3 20:26:53 GMT 2001
Set MOF2 to 25.0 as requested by Jonathan for a test.
**ECHO PROBLEM**: Wed Jan 3 21:06:57 GMT 2001
Doing the "Partial System Shutdown Sequence" at Kim's request, in an attempt
to clear some strange readings of QWP and MOF2 by software.
**ECHO PROBLEM**: Wed Jan 3 21:40:22 GMT 2001
Back up and taking data after that Partial Shutdown, MOF2 temp setpoint I set to 25.0, all else I left at defaults.
Wed Jan 3 22:08:19 GMT 2001
                            PICS
                                     End Patrol
Wed Jan 3 22:10:09 GMT 2001
                                     End Patrol
                             CHIP
                                     FrontEnd finished.
Wed Jan 3 22:11:54 GMT 2001
                             CHIP
Wed Jan 3 22:26:08 GMT 2001
       MkIV
18 23.rawmk4 18 35.rawmk4 18 47.rawmk4 18 59.rawmk4
18 26.rawmk4 18 38.rawmk4 18 50.rawmk4 19 02.rawmk4
```